Author's response to reviews

Title: Quality of life in Patients treated with first line antiretroviral therapy containing nevirapine or efavirenz in Uganda: a prospective non randomized study.

Authors:

Doris Mutabazi Mwesigire (dorismutabazi@yahoo.com)
Albert W Wu (awu@jhu.edu)
Faith Martin (psmfm@bath.ac.uk)
Janet Seeley (janet.seeley@mrcuganda.co.ug)
Achilles Katamba (akatamba@yahoo.com)

Version: 2  Date: 2 February 2015

Author's response to reviews:

We have revised the manuscript and added the line and numbers in the main text of the manuscript and a list of abbreviations preceding competing interests.